Author's response to reviews

Title: BRAF mutations in thyroid tumors from an ethnically diverse group

Authors:

Hans-Juergen Schulten (hschulten@kau.edu.sa)
Sherine Salama (salama.cherry@gmail.com)
Zuhoor Al-Mansouri (z-almansouri@hotmail.com)
Reem Alotibi (ralotibi@hotmail.com)
Khalid Al-Ghamdi (kbalghamdy@kau.edu.sa)
Osman Abdel Al-Hamour (osmanahamour@gmail.com)
Hasan Sayadi (sayadih@hotmail.com)
Hosam Al-Aradati (HAI-Aradati@kfshrc.edu.sa)
Adel Al-Johari (msa1959@gmail.com)
Mohammed Hussain Al-Qahtani (mhalqahtani@kau.edu.sa)
Jaudah Al-Maghrabi (jalmaghrabi@hotmail.com)

Version: 2 Date: 3 August 2012

Author's response to reviews:

Dear Editors-in-Chief,

The manuscript entitled

BRAF mutations in thyroid tumors from an ethnically diverse group

was under consideration of BMC Cancer and the Editorial Team of this journal suggested a transfer of the submission to Hereditary Cancer in Clinical Practice.

Best regards,

Hans-Juergen Schulten, PhD